Author's response to reviews

Title: Maternal and neonatal outcomes for an urban Indigenous population compared with their non-Indigenous counterparts and a trend analysis over three triennia.

Authors:

Sue Kildea (sue.kildea@acu.edu.au)
Helen Stapleton (helen.stapleton@acu.edu.au)
Rebecca Murphy (murphy30@tcd.ie)
Kristen Gibbons (kgibbons@mmri.mater.org.au)

Version: 2 Date: 28 June 2012

Author's response to reviews:

The following sentence was added to the methodology section: This study was approved by the ethics committee of Mater Hospital.

A conclusion heading had been omitted from the first submission. The conclusion heading has now been inserted just before the last two paragraphs of the manuscript.